Author's response to reviews

Title: The value of age and medical history for predicting colorectal cancer and adenomas in people referred for colonoscopy.

Authors:

Barbara-Ann Adelstein (b.adelstein@unsw.edu.au)
Petra Macaskill (petra.macaskill@sydney.edu.au)
Robin M Turner (robin.turner@sydney.edu.au)
Peter H Katelaris (peter.katelaris@sydney.edu.au)
Les Irwig (les.irwig@sydney.edu.au)

Version: 7 Date: 31 August 2011

Author's response to reviews:

Dear Editor

Thank you for the opportunity for final proofreading. There are several mistakes that I have rectified in the attached document. These are

Background: First paragraph, line 4: “adenomatous” has been deleted.
Page 7, first paragraph, last line: reference group has been changed to referent group.
Page 12, second paragraph line 2: replace , (comma) with . (full stop) at end of sentence.
Page 13 second paragraph, line 1: replace C with c (curve)

Yours sincerely

Barbara-Ann Adelstein